Author’s response to reviews

Title: The relationship between the level of μ- opioid receptor (μORs) and postoperative analgesic use in patients undergoing septoplasty: A prospective randomized controlled trial.

Authors:

MUZAFFER GENCER (dr.m.gencer07@gmail.com)
Ayşe Yeşim Göçmen (yesimgocmen@hotmail.com)

Version: 1 Date: 05 Mar 2020

Author’s response to reviews:

1. We note that the current submission contains some textual overlap with other previously published works, in particular:

'The Efficacy of Anterior and Posterior Archs Suturation at Inferior Tonsillar Pole for Posttonsillectomy Pain Control' https://doi.org/10.1007/s12070-011-0357-9
Overlap occurs in paragraph 5 of the Discussion.

While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. Please re-phrase these sections to minimise overlap.

2. In accordance with BioMed Central editorial policies (http://www.biomedcentral.com/submissions/editorial-policies#standards+of+reporting), could you please ensure your manuscript reporting adheres to CONSORT guidelines (http://www.consort-statement.org/) for reporting trials. This is so your methodology can be fully evaluated and utilized.

Can you please include a completed CONSORT checklist as an additional file when submitting your revised manuscript.

3. At this stage, please upload your manuscript as a single, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.